Author’s response to reviews

Title: ICU-treated influenza A(H1N1) pdm09 infections more severe post pandemic than during 2009 pandemic: a retrospective analysis

Authors:

Pekka Ylipalosaari (pekka.ylipalosaari@ppshp.fi; pylipalo@gmail.com)
Tero Ala-Kokko (tero.ala-kokko@ppshp.fi)
Jouko Laurila (jouko.laurila@ppshp.fi)
Lauri Ahvenjärvi (lauri.ahvenjarvi@ppshp.fi)
Hannu Syrjälä (hannu.syrjala@ppshp.fi)

Version: 2 Date: 25 Sep 2017

Author’s response to reviews:

Your manuscript "ICU-treated influenza A(H1N1) pdm09 infections more severe post pandemic than during 2009 pandemic: a retrospective analysis" (INFD-D-17-00367R1) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Infectious Diseases, once you have carried out some essential revisions:

--------

1. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Figure legends are now presented in separate sheet

2. Thank you for providing a statement regarding the Availability of data and materials. However, we cannot currently accept the information provided as it does not meet our editorial policies (see http://www.biomedcentral.com/getpublished/editorial-policies#DataandMaterialRelease).

Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality.

Please detail where the raw data supporting your findings can be found (including information found in any supplementary files). If the raw data is publicly available or can be requested,
please state that this is the case, and where the data can be found/requested from. Alternatively, if you do not wish to/cannot share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason.

Please note that participant or patient data should be de-identified.

Please see our submission guidelines for the statements we can accept in this section.

The data can be requested from dr. Pekka Ylipalosaari (email: pekka.ylipalosaari@ppshp.fi)

3. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of author LA is not clear, and does not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their name from the list of authors and place them in the “Acknowledgements” section instead. Please note that any changes in the list of authors requires the completion of the “Change in authorship” form.

Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group - alone - does not usually justify authorship.

The following sentence was added: L A re-evaluated all imaging results and he critically evaluated the final version of the manuscript.

4. Please acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

If there are no acknowledgements, please state “Not applicable”.

“Not applicable”

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

--------

Once you have made the necessary corrections, please submit a revised manuscript online at:
If you have forgotten your username or password please use the "Send Login Details" link to get your login information. For security reasons, your password will be reset.

We request that a point-by-point response letter accompanies your revised manuscript. This letter must provide a detailed response to each reviewer/editorial point raised, describing what amendments have been made to the manuscript text and where these can be found (e.g. Methods section, line 12, page 5). If you disagree with any comments raised, please provide a detailed rebuttal to help explain and justify your decision.

Please also ensure that your revised manuscript conforms to the journal style, which can be found in the Instructions for Authors on the journal homepage.

A decision will be made once we have received your revised manuscript, which we expect by 06 Oct 2017.

Please note that you will not be able to add, remove, or change the order of authors once the editor has accepted your manuscript for publication. Any proposed changes to the authorship must be requested during peer-review, and adhere to our criteria for authorship as outlined in BioMed Central's policies. To request a change in authorship, please download the 'Request for change in authorship form' which can be found here - http://www.biomedcentral.com/about/editorialpolicies#authorship. Please note that incomplete forms will be rejected. Your request will be taken into consideration by the editor, and you will be advised whether any changes will be permitted. Please be aware that we may investigate, or ask your institute to investigate, any unauthorized attempts to change authorship or discrepancies in authorship between the submitted and revised versions of your manuscript.